Author's response to reviews

Title: In vitro evaluation of novel N-acetylalaninate prodrugs that selectively induce apoptosis in prostate cancer cells

Authors:
Christopher A McGoldrick (cmcgoldr@uthsc.edu)
Yu-Lin Jiang (jiayu@musc.edu)
Marianne Brannon (leverone@etsu.edu)
Koyamangalath Krishnan (krishnak@etsu.edu)
William L Stone (stone@etsu.edu)

Version: 4 Date: 10 September 2014

Author's response to reviews: see over
We very much appreciate the constructive editorial revision and believe that our revision has improved the quality of this submission.

**Comment 1:** Please include a conclusion section.

**Response:** We have provided a conclusion section in the abstract without going beyond the word count limit.

There were no other comments or suggestions.